Response to reviewer and editor comments

We sincerely thank the reviewer for very constructive and encouraging comments, and for acknowledging the quality of our article.

In addition, we have addressed the comments and notes of the editor in the edited manuscript file, and modified our manuscript accordingly. We have also included one figure in response to the request by the Editor-in-Chief.